Author's response to reviews

Title: Vitamin C-induced hyperoxaluria causing reversible tubulointerstitial nephritis and chronic renal failure: a case report

Authors:

  Shradha Rathi (shradha-rathi@ouhsc.edu)
  William Kern (bill-kern@ouhsc.edu)
  Kai Lau (kai-lau@ouhsc.edu)

Version: 3 Date: 17 November 2007

Author's response to reviews: see over
Dear Editor:

We thank you for accepting our manuscript titled “Vitamin C-induced hyperoxaluria causing reversible tubulointerstitial nephritis and chronic renal failure: a case report” for publication. We would like to submit a revision with the following changes in the format as requested:

Formatting changes requested:

Affiliations: Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. Can we also ask you to link each author to their corresponding affiliation with the use of superscript numbers.

Done

Background - Please rename this section “Introduction”

Done

Competing Interests -> Competing interests

Done

Please put the consent statement under a section entitled 'Consent'

Done

Thank you for the opportunity to re-submit to the Journal.

Sincerely,

Kai Lau, M.D.
Professor of Medicine
Section of Nephrology
University of Oklahoma Health Sciences Center
Williams Pavilion, Room 2250
940 SL Young Blvd.,
Oklahoma City, OK 73104
United States
Phone: (405) 271-6842
Email: kai-lau@ouhsc.edu